Author's response to reviews

Title: Diagnosis of invasive aspergillus tracheobronchitis facilitated by endobronchial ultrasound-guided transbronchial needle aspiration: a case report

Authors:

Roberto F Casal (rfcasal@gmail.com)
Roberto Adachi (radachi@mdanderson.org)
Carlos A Jimenez (cajimenez@mdanderson.org)
Mona Sarkiss (msarkiss@mdanderson.org)
Rodolfo C Morice (rmorice@mdanderson.org)
Georgie A Eapen (geapen@mdanderson.org)

Version: 2 Date: 17 November 2009

Author's response to reviews:

Changes made to manuscript:
1. Ethnic origin and nationality of the patient were introduced.
2. Figures were removed from manuscript
3. Figure legends were left after the references
4. The word "address" was removed from title page
5. “Consent section” was placed before the “competing interests” section.